Author's response to reviews

Title: Violence witnessing, perpetrating and victimization in Medellin, Colombia. A random population survey

Authors:

Luis F Duque Prof (lfduque.ramirez@gmail.com)
Nilton E Montoya Mr (nilton_montoya@hotmail.com)
Alexandra Restrepo Dr (cheres80@gmail.com)

Version: 3 Date: 22 July 2011

Author's response to reviews: see over
Medellin, July 20, 2011

The BMC PUBLIC HEALTH JOURNAL

Dear sirs:

Please find attached the revised article *Violence witnessing, perpetrating and victimization in Medellin, Colombia: a random population survey*. Authors thank very much to reviewers for their inputs and suggestions, which we accepted and were the basis for this revision.

In order to facilitate reviewers work we are sending this article under the changes control mode.

Please do not hesitate to ask any other aspects you or reviewers might consider necessary.

Sincerely,

Luis F Duque, Nilton E. Montoya, Alexandra Restrepo